Author's response to reviews

Title: Comparison of diabetes management in five countries for general and indigenous populations: an internet-based review

Authors:

Damin Si (damin.si@menzies.edu.au)
Ross Bailie (ross.bailie@menzies.edu.au)
Zhiqiang Wang (z.wang@uq.edu.au)
Tarun Weeramanthri (tarun.weeramanthri@health.wa.gov.au)

Version: 5 Date: 16 June 2010

Author's response to reviews: see over
Dear Editor,

We have made formatting changes as requested, as outlined below:

**Major revisions**

*References*: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Currently your references are cited out of order. Please correct this.

We have fixed the problem related to the sequence of references in the 1st and 2nd paragraphs and in Table 1. Now, reference numbers are consecutive in the full text of the manuscript.

*Typography*: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have proofread the manuscript to correct errors identified.

Thank you again for your help and advice.

Best wishes,

Damin

Dr Damin Si